

# 

**Citation:** Luvai EAC, Kyaw AK, Sabin NS, Yu F, Hmone SW, Thant KZ, et al. (2021) Evidence of Chikungunya virus seroprevalence in Myanmar among dengue-suspected patients and healthy volunteers in 2013, 2015, and 2018. PLoS Negl Trop Dis 15(12): e0009961. https://doi.org/ 10.1371/journal.pntd.0009961

Editor: Kate Zinszer, Universite de Montreal, CANADA

Received: March 29, 2021

Accepted: November 1, 2021

Published: December 1, 2021

**Copyright:** © 2021 Luvai et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This research was supported by the Japan Agency of Medical Research and Development (AMED) (https://www.amed.go.jp) through Japan Program for Infectious Diseases Research and Infrastructure, under grant numbers JP20wm0125006 and JP21wm0225017 to coinvestigators KM and SI respectively. The funders RESEARCH ARTICLE

# Evidence of Chikungunya virus seroprevalence in Myanmar among denguesuspected patients and healthy volunteers in 2013, 2015, and 2018

Elizabeth Ajema Chebichi Luvai<sup>1,2,3</sup>, Aung Kyaw Kyaw<sup>6</sup><sup>4</sup>, Nundu Sabiti Sabin<sup>2</sup>, Fuxun Yu<sup>5</sup>, Saw Wut Hmone<sup>6</sup>, Kyaw Zin Thant<sup>4</sup>, Shingo Inoue<sup>1</sup>, Kouichi Morita<sup>1,2\*</sup>, Mya Myat Ngwe Tun<sup>1\*</sup>

Department of Virology, Institute of Tropical Medicine (NEKKEN), Nagasaki University, Nagasaki, Japan,
Program for Nurturing Global Leaders in Tropical and Emerging Communicable Diseases, Graduate
School of Biomedical Sciences, Nagasaki University, Nagasaki, Japan, 3 Department of Biomedical
Sciences and Technology, School of Health and Biomedical Sciences, The Technical University of Kenya,
Nairobi, Kenya, 4 Department of Medical Research, Ministry of Health and Sports, Yangon, Myanmar,
Guizhou Provincial People's Hospital, Guiyang City, Guizhou Province, China, 6 Department of Pathology,
University of Medicine-1, Lanmadaw township, Yangon, Myanmar

\* moritak@nagasaki-u.ac.jp (KM); myamyat@tm.nagasaki-u.ac.jp (MMNT)

# Abstract

# Introduction

Chikungunya virus (CHIKV) is a mosquito-borne virus known to cause acute febrile illness associated with debilitating polyarthritis. In 2019, several institutions in Myanmar reported a CHIKV outbreak. There are no official reports of CHIKV cases between 2011 and 2018. Therefore, this study sought to determine the seroprevalence of CHIKV infection before the 2019 outbreak.

# Methods

A total of 1,544 serum samples were collected from healthy volunteers and patients with febrile illnesses in Yangon, Mandalay, and the Myeik district in 2013, 2015, and 2018. Participants ranged from one month to 65 years of age. Antibody screening was performed with in-house anti-CHIKV IgG and IgM ELISA. A neutralization assay was used as a confirmatory test.

# Results

The seroprevalence of anti-CHIKV IgM and anti-CHIKV IgG was 8.9% and 28.6%, respectively, with an overall seropositivity rate of 34.5%. A focus reduction neutralization assay confirmed 32.5% seroprevalence of CHIKV in the study population. Age, health status, and region were significantly associated with neutralizing antibodies (NAbs) and CHIKV seropositivity (p < 0.05), while gender was not (p = 0.9). Seroprevalence in 2013, 2015, and

had no role in the study design, data collection, and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

2018 was 32.1%, 28.8%, and 37.3%, respectively. Of the clinical symptoms observed in participants with fevers, arthralgia was mainly noted in CHIKV-seropositive patients.

### Conclusion

The findings in this study reveal the circulation of CHIKV in Myanmar's Mandalay, Yangon, and Myeik regions before the 2019 CHIKV outbreak. As no treatment or vaccine for CHIKV exists, the virus must be monitored through systematic surveillance in Myanmar.

#### Author summary

Few CHIKV outbreaks have been detected in Myanmar since the first documented case in 1973. After an outbreak, the virus seems to disappear from the region gradually for a few years to more than a decade. In 2019, a CHIKV outbreak was reported in blood donors and children with febrile illness in the Mandalay region. The last official report of a CHIKV outbreak before this was in 2010. Our findings showed evidence of both IgG (28.6%) and IgM (8.9%) antibody circulation against CHIKV. In 2018, the highest sero-prevalence rate (37.3%) was found, a probable predictor of the CHIKV outbreak reported in 2019. Additionally, we observed an overall prevalence of 32.5% of circulating anti-CHIKV neutralizing antibodies in the study population. Neutralizing antibodies were observed in patients with febrile illness and healthy volunteers. These findings indicate a continued risk for future outbreaks, reinforcing the need to monitor potential outbreaks in the country.

# Introduction

Chikungunya virus (CHIKV) is an alphavirus in the *Togaviridae* family [1]. The virus is classified as an arthropod-borne virus (arbovirus) transmitted primarily by *Aedes aegypti* and *A*. *albopictus* mosquitoes, which are endemic in tropical and subtropical regions [2–4]. The clinical presentation of CHIKV disease varies from self-limiting undifferentiated febrile illness to debilitating polyarthritis and encephalitis and, in some cases, death may occur [5, 6]. According to the World Health Organization (WHO), CHIKV is an emerging public health threat worldwide [7].

CHIKV was first documented in 1952 in Tanzania [8]. A major outbreak reported in Kenya in 2004 [9] led to the spread of CHIKV to the islands of the Indian Ocean, India, and Southeast Asia, with millions of reported cases [10]. CHIKV arrived in the Americas in 2013 and spread from the Caribbean islands to Brazil in 2014 [11, 12]. Imported cases have been documented in various countries in Europe, North America, East Asia, and the Middle East [1, 13–16] and autochthonous cases have been reported in France and Italy [17, 18].

In Myanmar, the first case of CHIKV occurred in 1973 [19], with subsequent cases reported in 1998, 2006, 2008, and 2010 [20–22]. In 2019, a newspaper stated that the Department of Public Health in Myanmar had identified an outbreak of CHIKV in Kachin State, Nay Pyi Taw, and the Tanintharyi region [23]. In the same year, the GeoSentinel Surveillance Network reported 18 cases of CHIKV infection in travelers returning from Myanmar [15]. Additionally, in 2019, an outbreak of CHIKV infection was detected in 20.5% of children with acute febrile illness and 3.2% of blood donors in the Mandalay region [24]. According to reports, after a large-scale CHIKV outbreak in Myanmar, the virus seems to disappear from the region gradually for a few years to more than a decade [20, 23, 25]. However, countries neighboring Myanmar, such as India, frequently report consistent infection rates of CHIKV [20]. It is unclear what factors trigger multiple institutions' detection of the virus in the regions where outbreaks occur. CHIKV is transmitted by similar vectors as dengue virus (DENV) and, during the early stages of infection, their primary symptoms are indistinguishable [26–29]. DENV and CHIKV coinfection has been reported in Asia, Africa, and some parts of the Americas [28, 30–33]. The co-circulation and clinical similarities of these arboviruses, as well as the limited capacity for CHIKV testing, have contributed to the under-diagnosis of CHIKV in the regions of Myanmar where the two viruses co-circulate [34, 35].

In Myanmar, there is a lack of active screening and surveillance for CHIKV infection. There were no official reports of CHIKV cases between 2010 and 2019 [15]. Therefore, this study aimed to determine the seroprevalence of CHIKV in healthy volunteers and patients suspected to have dengue fever in 2013, 2015, and 2018.

# Methods

#### Ethics statement

Ethical approvals for this study were obtained from the Institutional Ethical Committee on Medical Research Involving Human Subjects in Myanmar (1/2012, 6/2015, 097/2017, and 082/2018) and the Institute of Tropical Medicine Ethical Committee, Nagasaki University, Japan (171207186–2, 191003223, and 200619241). Before sample collection, written informed consent was obtained from both patients and healthy donors and the parents or legal guardians of participating children.

# Samples and study population

This study retrospectively analyzed serum samples that had been used in previous research studies [36–38]. The study population comprised 1,544 healthy volunteers and febrile patients with suspected DENV infection. The study participants were from three distinct regions: Mandalay, Yangon, and Myeik (S1 Fig). Mandalay is in the upper Myanmar zone and borders India, China, and Bangladesh. Yangon is located in southern Myanmar and is the largest city in the country. Myeik is a coastal region, part of the Mergui archipelago in the extreme south of Myanmar that borders Thailand.

The healthy volunteers consisted of 934 observably healthy and asymptomatic individuals with no history of hospitalization for at least six months before sample collection. Of the 934 individuals, 421 were selected from three monastic schools in Mandalay, and 513 were selected from private clinics in Yangon during routine medical examinations. All of the samples from healthy volunteers were collected in 2018.

There were approximately 610 individuals suspected of dengue fever, of whom, 104 were from Myeik and 506 were from Mandalay. The samples from symptomatic people were collected in 2013 and 2015 to screen for DENV infection during outbreaks.

# Viruses and cell lines

The CHIKV strain S-27, African prototype, was used for CHIKV IgG indirect ELISA, CHIKV IgM capture ELISA, and neutralization testing. The virus was propagated in C636 mosquito cells and used for viral titration. Vero cells (African green monkey kidney epithelial cells, ATCC CCL-81) were used for neutralization tests.

### Detection of anti-CHIKV IgG

The detection of anti-CHIKV IgG was performed to determine progressive or past CHIKV infection in the populations. To screen for anti-CHIKV IgG in the serum samples, an in-house indirect IgG ELISA was performed using purified CHIKV as the assay antigen [39]. Detection of IgG antibodies was carried out following the procedure described in previous studies [21, 22, 24, 40] with minor adjustments. Briefly, 96-well microplates (Nalge Nunc International, Denmark) were coated with antigen (125 ng/100 uL per well) and diluted in coating buffer, except for the blank wells. The plate was incubated overnight at 4°C. The test samples and positive and negative controls were diluted in 1:1000 phosphate-buffered saline in Tween 20 (PBS-T) in 10% Block Ace (Yukijirushi, Japan) and were distributed into duplicate wells. Subsequently, 1:25,000 diluted horseradish peroxidase (HRP) conjugated anti-human IgG (American Qualex, USA) in PBS-T with 10% Block Ace was added. The color was developed by adding o-phenylenediamine dihydrochloride solution (OPD; Sigma Chemical Co, USA) to each well with 0.03% hydrogen peroxide in 0.05 M citrate phosphate buffer (pH 5.0). After incubation for an hour at room temperature, the reaction was stopped with 1N sulfuric acid (1N H<sub>2</sub>SO4), and the optical density was read at 492 nm with Multiscan JX. The IgG titers of patients' serum samples were determined from the positive standard curve. A sample titer of  $\geq$  3000 was considered IgG positive. To validate the in-house IgG indirect ELISA, 300 samples were screened, and a 50% focus reduction neutralization assay (FRNT<sub>50</sub>) was utilized as the standard.

#### Detection of IgM antibodies

Tests for anti-CHIKV IgM were performed to identify recent CHIKV infections in the population. The presence of anti-CHIKV IgM was detected with an in-house IgM capture ELISA system [21, 22, 24, 40]. All wells, except the blank wells, were coated with anti-human IgM goat IgG (Cappel ICN Pharmaceuticals, USA) in 0.05 M carbonate-bicarbonate buffer (pH 9.6) containing 0.02% sodium azide as the diluent. After overnight incubation and blocking, the test samples and positive and negative controls diluted in PBS-T with 10% Block Ace were distributed into duplicate wells. Subsequently, 128 ELISA units of CHIKV assay antigen (strain S-27, African prototype) were added to each well and incubated for 1 hour at 37°C. A dilution of 1:400 of HRP conjugated anti-CHIKV mouse-derived recombinant E1 monoclonal antibody was added and incubated for 1 hour at 37°C. The reaction color was developed and the optical density reading was performed as previously described for IgG detection. A positive control OD<sub>492</sub>/negative control OD<sub>492</sub> (P/N) ratio  $\geq$  2.0 was considered positive. To validate the inhouse IgM system, we analyzed randomly selected samples with the human anti-CHIKV Abcam IgM ELISA Kit (ab177848) that has been utilized in other studies as the standard [41, 42].

#### Neutralizing assay

The neutralizing activity of antibodies from the IgG- and IgM-positive sera was confirmed with a 50% focus reduction neutralization test (FRNT<sub>50</sub>) as described in previous studies [21, 40]. The heat-treated serum samples were mixed with equal volumes of 40 focus-forming units. After incubation at 37 °C for 1 hour, the mixture was transferred to duplicate 96-well plates of confluent Vero cell monolayers. After incubation at 37 °C for 1.5 hours, the cells were overlaid with 150  $\mu$ L of 2% FCS MEM containing 1% methylcellulose 4000 (WAKO Pure Chemical Industries, Japan). The plates were then incubated at 37 °C with 5% CO<sub>2</sub> for 36 hours. After fixing the cells, they were blocked and permeabilized as described in previous studies [21]. Viral foci were detected by immunostaining the cells with anti-CHIKV serum

from C57BL/6J mice, peroxidase-conjugated anti-mouse IgG (American Qualex, USA), and DAB substrate (WAKO Pure Chemical Industries, Japan). The endpoint serum dilution that provided a  $\geq$  50% reduction compared with the mean number of the control well was considered the FRNT<sub>50</sub> titer. Confirmed CHIKV cases were defined as IgG- or IgM-positive with a neutralization titer of  $\geq$  10.

#### Statistical analysis

Data analysis was performed with GraphPad Prism 9.0.1 (GraphPad Software) and Stata Corp (2019) Stata Statistical Software: Release 16 (StataCorp LLC, College Station, TX).

Chi-square tests were used to determine the differences in proportions of risk factors among groups. An initial analysis produced univariate odds ratio (OR) estimates (95% confidence interval) for the potential risk factors using logistic regression, followed by a multivariable logistic regression model to adjust the OR. Additionally, average marginal effects (AMEs) were calculated. Akaike's information criterion (AIC) and Bayesian information criteria (BIC) were used to select the best model for the study [43–45]. The Kruskal-Wallis H test and Mann-Whitney U test were used to determine the difference in median among groups. The correlation between anti-CHIKV neutralizing antibodies and IgG or IgM seropositivity was determined with Spearman's correlation coefficient *r*. All test results were considered statistically significant at p < 0.05.

#### Results

#### **Demographic characteristics**

The demographic characteristics of the study participants are summarized in **Table 1**. The study population comprised a total of 1,544 participants, 39.0% (602/1,544) female and 61.0% (942/1,544) male. The median age was 12 and the interquartile range was 7–23 years old. The participants were divided into four age groups based on the risk of acquiring severe CHIKV

Variable	Overall number (%)	Female (%)	Male (%)
Age* (years)			
≤5	268 (17.4)	134 (50.0)	134 (50.0)
6-15	701 (45.4)	328 (47.8)	373 (53.2)
16-45	554(35.9)	134 (24.2	420(75.8)
$\geq 46$	21 (1.3)	6 (28.6)	15 (71.4)
Region			
Yangon	513 (33.2)	127 (24.7)	386 (75.2)
Mandalay	927 (60.0)	425 (42.9)	502 (54.1)
Myeik	104 (6.8)	50 (48.0)	54 (51.9)
Health status			
Febrile patients	610 (39.5)	302 (49.5)	308 (50.5)
Healthy volunteers	934 (60.5)	300 (32.1)	634 (67.9)
Year of collection			
2013	280 (18.1)	133 (47.5)	147 (52.5)
2015	330 (21.4)	169 (51.2)	161 (48.8)
2018	934 (60.5)	300 (32.1)	634 (67.9)

Table 1. Demographic characteristics of the study population (N = 1,544).

\* The participants' ages were defined based on the date and year of sample collection.

disease [46–50]:  $\leq$  5 years old, 6–15 years old, 16–45 years old, and  $\geq$  46 years old. Most of the participants (45.4%) were school-aged children (6–15 years old). The Mandalay region had the highest proportion of participants (60.0%), whereas Myeik had the lowest (6.8%). The study participants were also grouped by health status: healthy volunteers represented 60.5% of participants and febrile-illness patients, 39.5%. Most of the study participants (60.5%) were sampled in 2018, followed by 21.4% in 2015 and 18.1% in 2013.

# CHIKV IgG and IgM seroprevalence in the study population

The validation results showed that the sensitivity and specificity of the in-house anti-CHIKV IgM capture ELISA were 98.3% and 88%, respectively (S1 Table). Additionally, the sensitivity and specificity of the anti-CHIKV IgG indirect ELISA were 94.2% and 100%, respectively (S2 Table).

Of the 1,544 serum samples tested, 28.6% were positive for anti-CHIKV IgG and 8.9% were positive for anti-CHIKV IgM (Table 2). Overall, the seroprevalence of CHIKV IgG and IgM antibodies was 34.5% (Table 2). Approximately 3.0% of the study population had both IgG and IgM antibodies. Despite representing the smallest group of participants in the study, the older population ( $\geq$  46 years) had the highest seroprevalence (85.7%), followed by the 16–45-year-old group (44.4%). The 6–15-year-old group and the children  $\leq$  5 years old had seroprevalences of 29.2% and 23.9%, respectively. Participants from Yangon exhibited the highest seroprevalence rate (47.9%), followed by those from Myeik (42.3%), and those from Mandalay had the lowest seroprevalence at 26.2%. The seroprevalence rate was higher in males (36.2%) compared with females (31.9%). Notably, the overall seroprevalence rate was lower in the febrile-patient population (30.3%) than among the healthy volunteers (37.3%). Samples taken in 2018 showed the highest seropositivity (37.3%), followed by those from 2013 (32.1%) and 2015 (28.8%) (Table 2).

Table 2.	Anti-CHIKV	seropositivity	rate in t	the study	population	(N = 1,544).
----------	------------	----------------	-----------	-----------	------------	--------------

Variable	Number	IgG positive (%)	IgM positive (%)	IgG and IgM positive (%)	IgG and/or IgM positive (%)
Age (years)					
<u>≤</u> 5	268	40 (14.9)	27 (10.1)	3 (7.5)	64 (23.9)
6-15	701	164 (23.4)	61 (8.7)	20 (12.2)	205 (29.2)
16-45	554	221 (39.9)	46 (8.3)	21 (9.5)	246 (44.4)
≥46	21	17 (80.9)	3 (14.3)	2 (11.8)	18 (85.7)
Region					
Mandalay	927	187 (20.2)	75 (8.1)	19 (10.2)	243 (26.2)
Myeik	104	33 (31.7)	19 (18.3)	8 (24.2	44 (42.3)
Yangon	513	222 (43.6)	43 (8.4)	19 (8.6)	246 (47.9)
Gender					
Female	602	154 (25.6)	51 (8.5)	13 (8.4)	192 (31.9)
Male	942	288 (30.5)	86 (9.2)	33 (11.5)	341 (36.2)
Health status					
Febrile patients	610	152 (24.9)	48 (7.9)	15 (9.7)	185 (30.3)
Healthy volunteers	934	290 (31.1)	89 (9.5)	31 (10.7)	348 (37.3)
Year					
2013	280	83 (29.6)	13 (4.6)	6 (7.2)	90 (32.1)
2015	330	69 (20.9)	35 (10.6)	9 (13.0)	95 (28.8)
2018	934	290 (31.1)	89 (9.5)	31 (10.7)	348 (37.3)
Overall	1544	442 (28.6)	137 (8.9)	46 (3.0)	533 (34.5)





https://doi.org/10.1371/journal.pntd.0009961.g001

Further analysis to determine the distribution of CHIKV antibodies among the seropositive population revealed that the amount of anti-CHIKV IgG increased with increasing age (**Fig 1A**). In contrast, anti-CHIKV IgM seroprevalence increased with decreasing age, with school-aged children (6–15 years) having the highest seroprevalence rate (10.2%) of individuals with both IgM and IgG CHIKV antibodies (**Fig 1A**). A high anti-CHIKV IgG seroprevalence rate was observed in samples taken from Mandalay (69.1%) and Yangon (80.2%), whereas Myeik had the highest distribution of anti-CHIKV IgM antibodies (25.0%) (**Fig 1B**).

The percentage of anti-CHIKV IgG and IgM and the presence of both IgG and IgM among the seropositive population by age group (A) and region (B) are indicated in the graphs.

The association of the independent variables (age, region, gender, and health status) with the IgM or IgG seropositivity rate is illustrated in Table 3. In the multivariable logistic

			Univariable	e analysis	Multivariable analysis	
Variable	Number	IgG and/or IgM positive (%)	cOR* (95% CI)	<i>p</i> -value	aOR* (95% CI)	<i>p</i> -value
Age (years)						
≤5	268	23.9	Ref.		Ref.	
6-15	701	29.2	1.3 (0.9–1.8)	0.096	1.7 (1.2–2.4)	0.004
16-45	554	44.4	2.5 (1.8–3.5)	<0.0001	1.9 (1.0–3.8)	0.05
≥46	21	85.7	19.1 (5.5–67.0)	<0.0001	13.4 (3.3–54.7)	<0.0001
Region						
Mandalay	927	26.2	Ref.		Ref.	
Myeik	104	42.3	2.1 (1.4-3.1)	0.001	1.9 (1.2–3.0)	0.003
Yangon	513	47.9	2.6 (2.1-3.3)	<0.001	2.4 (1.3-4.2)	0.004
Gender						
Female	602	31.9	Ref.		Ref.	
Male	942	36.2	1.2 (1.0–1.5)	0.083	1.0 (0.8–1.3)	0.8
Health status						
Healthy volunteers	934	37.3	Ref.		Ref.	
Febrile patients	610	30.3	1.4 (1.1–1.6)	0.005	1.5 (1.1–2.1)	0.02

Table 3. Association of CHIKV seroprevalence rate with age, region, gender, and health status (N = 1,544).

CHIKV: Chikungunya virus; cOR: crude odds ratio; aOR: adjusted odds ratio; CI: confidence interval. Seroprevalence rate includes both IgG and IgM positive results.

regression model, age, region, and health status were significantly associated with the seroprevalence rate (p < 0.05) Table 3. Using the multivariable logistic regression model, we found that children  $\geq$  6 years old were 1.7 times (95% CI: 1.2–2.4) more likely to be CHIKV-seropositive than children  $\leq$  5 years old, and individuals 16–45 years old were 1.9 times (95% CI: 1.0– 3.8) more likely to be CHIKV-seropositive than children < 5 years old. Individuals > 46 years old were 13.4 times (95% CI: 3.3-54.7) more likely to be CHIKV-seropositive compared to children  $\leq$  5 years old. The adjusted AMEs estimates of CHIKV seropositivity among the 6– 15-year-old, 16–45-year-old, and  $\geq$  46-year-old age groups were 0.11% (95% CI: 0.04%– 0.17%), 0.11% (95% CI: -0.02%-0.2%) and 0.53% (0.3%-0.8%), respectively, higher than children  $\leq$  5 years (S5 Table). Furthermore, individuals from Myeik were 1.9 times (95% CI: 1.2-3.0), and Yangon, 2.4 times (95% CI: 1.3-4.2) more likely to be CHIKV-seropositive compared with those from Mandalay (Table 3). Additionally, the adjusted AMEs estimates of CHIKV seropositivity were 0.14% (95% CI: 0.04%-0.2%) and 0.19% (95% CI: 0.06%-0.3%) higher in Myeik and Yangon, respectively, than in Mandalay (S5 Table). Notably, the febrile patients were 1.5 times (95% CI: 1.1-2.1) more likely to be CHIKV-seropositive than the healthy volunteers. The adjusted AMEs estimate of CHIKV seroprevalence among the febrile patients was 0.09% (95% CI: 0.02%-0.15%) higher than in healthy volunteers.

#### Seroprevalence of anti-CHIKV IgG and IgM in 2013, 2015, and 2018

The highest distribution of anti-CHIKV IgG antibodies was observed in the 2018 group (290/ 934), followed by 2013 (83/280) and 2015 (69/330), indicating a lower circulation of the antibodies in earlier years (**Fig 2A**). Samples from 2015 showed a high circulation of IgG with a lower titer, as indicated by the upper quartile range below the cutoff line (**Fig 2A**). Samples from 2013 had the lowest distribution of IgM antibodies (13/280), whereas those from 2015 (35/330) and 2018 (89/934) showed a fairly higher circulation of the antibodies in the study population (**Fig 2B**).

# Prevalence of anti-CHIKV neutralizing antibodies (NAbs) in the study population

All the IgM and IgG seropositive samples were confirmed CHIKV-positive by 50% focus reduction neutralizing titer (FRNT<sub>50</sub>). Overall, the anti-CHIKV neutralizing antibody



**Fig 2. Distribution of anti-CHIKV IgG and IgM antibodies in 2013, 2015, and 2018.** (A) An anti-CHIKV IgG titer of  $\geq$  3000 was considered CHIKVpositive and is indicated by a black dotted line. (B) An anti-CHIKV IgM-positive/negative (P/N) ratio of  $\geq$  2.0 was considered IgM-positive and is denoted by a black dotted line. Each dot represents the sample distribution of IgG and IgM in the study population, and the black error bars denote the median and interquartile ranges in figures A and B, respectively.



**Fig 3. Distribution of NAbs within the study population.** The presence of anti-CHIKV NAbs in the serum of participants was categorized according to age group (A) and region (B). Each dot above the whiskers indicates the outliers observed in each group. A Kruskal-Wallis H test was used to determine statistical significance among the age groups and regions, p < 0.0001 is denoted by \*\*\*\*.

https://doi.org/10.1371/journal.pntd.0009961.g003

prevalence rate was confirmed at 32.5% (502/1,544). The FRNT<sub>50</sub> titers ranged from 20– 10,240. The older age group ( $\geq$  46 years) had the highest mean NAb titer of 1,885 (95% CI: 498–3,272), however, the geometric mean antibody titer (GMT) was 806 (95% CI: 352–1,845) (**S7 Table**). The Yangon region had the highest mean NAb titer, 923 (95% CI: 756–1,091). The geometric mean antibody titer values (GMT) patterns were more or less similar to the mean NAb titers observed for the region, gender, health status, and year (**S7 Table**). The median NAb titers for the  $\leq$  5 years old age group were the lowest compared with the other three age groups (**Fig 3A**), which is consistent with the pattern of the mean NAb titer, with several samples above the 10<sup>4</sup> range (**Fig 3A**). The differences among each age group were statistically significant (p < 0.0001). Most high NAb titers were from the Yangon region, whereas Mandalay had NAbs with both the lowest and highest titers (**Fig 3B**). The NAb interquartile ranges and median titers of healthy volunteers were significantly higher than those in febrile patients (p = 0.007) (**Fig 4A**). Additionally, **Fig 4B** shows that a higher percentage of healthy volunteers had titers above 10<sup>3</sup>, whereas most febrile patients had much lower titers.

Further analysis was performed to determine the association between independent factors (age, region, gender, and health status) and CHIKV NAb prevalence (**Table 4**). As in the sero-positivity results, in the multivariable logistic regression model, age, region, and health status were significantly associated with NAb prevalence, while gender was not (p = 0.9). The multivariable logistic regression model indicated that the odds of CHIKV NAb prevalence were 1.7 times (95% CI: 1.2–2.6) and 2.1 times (95% CI: 1.1–4.2) higher among children  $\geq$  6 years old and adults 16–45 years old, respectively, than in children  $\leq$  5 years old (**Table 4**). Interestingly, the odds of CHIKV NAb prevalence were 16.1 times (95% CI: 4.0–65.7) higher in the  $\geq$  46-year-old group than in children  $\leq$  5 years old (**Table 4**). The adjusted AMEs estimates of CHIKV NAb prevalence among the 6–15-year-old, 16–45-year-old, and  $\geq$  46-year-old age groups were 0.11% (95% CI: 0.04%–0.17%), 0.15% (95% CI: 0.02%–0.28%), and 0.59%; (0.36%–0.81%), respectively; higher than in children  $\leq$  5 years old (**S6 Table**). Notably, the study participants from Yangon were 2.4 times (95% CI: 1.3–4.3), and Myeik, 1.9 times (95% CI: 1.2–2.9), more likely to have high CHIKV NAbs compared with those from Mandalay, as shown in **Table 4**. Additionally, the adjusted AMEs of CHIKV NAbs were 0.13% (95% CI:



**Fig 4. Distribution of CHIKV NAbs among healthy volunteers and febrile patients.** A) Distribution of NAbs was computed as a continuous approximation of the probability density using Kernel density estimation (KDE). The densities of both populations are annotated with the median (blue straight line) and the interquartile range (red dotted lines). The distribution of healthy volunteers' NAbs shows the median of samples above the  $10^3$  titer. The probability density of febrile patients is concentrated below the  $10^3$  NAb titer. Statistical significance was measured with the Mann-Whitney U test, p = 0.007 is denoted by \*\*\*. B) Comparison of the proportion of the NAb titer in healthy volunteers (blue) and febrile patients (red).

https://doi.org/10.1371/journal.pntd.0009961.g004

0.03%–0.23%) and 0.19% (95% CI: 0.06%–0.32%) higher in Myeik and Yangon, respectively, than in Mandalay (**S6 Table**). As previously observed in the seropositivity results, the prevalence of CHIKV NAb was 1.6 times (95% CI: 1.2–2.2) higher in febrile patients than in healthy volunteers (**Table 4**). The adjusted AMEs estimate of CHIKV NAb prevalence was 0.1% (95% CI: 0.03%–0.16%) higher in febrile patients than in healthy volunteers (**S6 Table**).

# Correlation between CHIKV IgG and IgM antibodies with NAbs

A higher number of anti-CHIKV IgG antibodies were able to neutralize the virus than anti-CHIKV IgM antibodies (Fig 5). The anti-CHIKV IgG titer and NAb titer were positively and

		NAbs (%)	Univariable analysis		Multivariable analysis	
Variable	Number		cOR*(95%CI)	<i>p</i> -value	aOR*(95%CI)	<i>p</i> -value
Age (years)						
<u>≤</u> 5	268	21.6	Ref.		Ref.	
6-15	701	27.3	1.4 (1.0–1.9)	0.075	1.7 (1.2–2.6)	0.002
16–45	554	42.4	2.6(1.9-3.7)	<0.0001	2.1 (1.1-4.2)	0.03
≥46	21	85.7	21.7 (6.2–76.3)	< 0.0001	16.1 (4.0-65.7)	0.0001
Region						
Mandalay	927	24.3	Ref.		Ref.	
Myeik	104	39.4	2.0 (1.3-3.1)	0.001	1.9 (1.2–2.9)	0.006
Yangon	513	46.0	2.7 (2.1-3.3)	<0.0001	2.4 (1.3-4.3)	0.004
Gender						
Female	602	29.9	Ref.		Ref.	
Male	942	34.2	1.2 (1.0–1.5)	0.08	1.0 (0.8–1.3)	0.8
Health status						
Healthy volunteers	934	35.1	Ref.		Ref.	
Febrile patients	610	28.5	0.7 (0.6–0.9)	0.007	1.6 (1.2–2.2)	0.005

Table 4. Association of the anti-CHIKV NAbs with age, region, gender, and health status (N = 1,544).

NAbs: Neutralizing antibodies; cOR: crude odds ratio; aOR: adjusted odds ratio; CI: confidence interval.



Fig 5. Comparison between CHIKV antibody titer and NAbs. https://doi.org/10.1371/journal.pntd.0009961.g005

significantly correlated (r = 0.8; 95% CI: 0.7–0.8, p < 0.0001) as shown in S2A Fig. Notably, the correlation between anti-CHIKV IgM titer and NAb titer (r = 0.05; 95% CI: 0.002–0.1, p = 0.04) was weaker, as shown in S2B Fig. Anti-CHIKV IgG antibodies had a higher NAb titer than IgM antibodies, illustrated by the geometric mean titer (with 95% CI bars) given in Fig 5. Approximately 440/442 anti-CHIKV IgG antibodies were able to neutralize the virus, compared with 105/137 of the anti-CHIKV IgM antibodies (Fig 5).

Sero-neutralization was performed on sera diluted at 1:10 to detect neutralizing IgM and IgG. The presence and absence of neutralizing titers (NT) of IgM and IgG were denoted by NT+ and NT-, respectively. The black bars indicate the geometric mean titers and 95% CI for both IgG and IgM.

# Presence of anti-CHIKV and anti-DENV antibodies in the febrile patient population

To further compare CHIKV infection in febrile patients in this study population, data from previous studies were utilized to evaluate the presence of DENV in the same population [36, 37]. The flow chart (S3 Fig) shows that, of the 610 patients, 30.3% were CHIKV-seropositive, and 5.9% of these had only CHIKV antibodies while 24.4% had both DENV and CHIKV antibodies. Upon evaluation of the specific signs and symptoms that were observed in the febrile patient population, arthralgia was observed in CHIKV cases and dual DENV- and CHIKV-seropositive individuals (S4 Fig).

# Discussion

CHIKV infection has become a major public health concern globally and poses a significant socio-economic burden. The findings in this paper confirm the circulation of CHIKV in 2013, 2015, and 2018, and one-third of the study participants were confirmed CHIKV-seropositive.

We demonstrated the occurrence of CHIKV infection among dengue-suspected patients and healthy volunteers in three distinct regions in Myanmar. This study revealed that 28.6% of the participants had anti-CHIKV IgG antibodies, 8.9% had anti-CHIKV IgM antibodies, and 3.0% had both IgM and IgG anti-CHIKV antibodies, with 34.5% overall seropositivity.

The study revealed the highest seroprevalence rate (37.3%) in 2018, a probable predictor of the outbreak reported in 2019 [15, 23, 24]. Around the same period, neighboring countries, such as Thailand and India, reported CHIKV seroprevalence rates of 26.8% and 18.1%, respectively [51, 52]. Interestingly, our study revealed that 2013 also had a high seroprevalence rate (32.1%), however, IgM seroprevalence was lower compared with the other tested years. High anti-CHIKV IgG seroprevalence in 2013 might be attributed to past infections. In contrast, 2015 showed the highest CHIKV IgM antibody circulation (10.7%), indicating infection within that year. The high seroprevalence rate could be attributed to cross-border spread as neighboring countries, such as Laos and Bangladesh, reported high CHIKV seroprevalence (43–90%) around the same period in which our samples were collected [53, 54].

In 2010, a study reported that the seroprevalence of CHIKV in Mandalay was 5.2% [21]. Our findings show an increase to 26.2% in that region. By using an adjusted odds ratio (aOR), we determined that participants from Myeik (aOR: 1.9, 95% CI: 1.2–3.0) and Yangon (aOR: 2.4, 95% CI: 1.3–4.2) are approximately two times more at risk of CHIKV infection than those from Mandalay. The reason the CHIKV seropositivity rate in Mandalay is lower than the other two cities, despite previous reports of the high endemicity of other arboviruses transmitted by similar vectors, is unclear [36–38, 55]. Both IgG and IgM seropositivity were observed at all the study sites, indicating ongoing infection despite previous exposure, and also suggesting that CHIKV endemicity is maintained within the studied regions [56].

This study's findings revealed that CHIKV seroprevalence increased with age, from 23.9% among children < 5 years old to 29.2% (aOR: 1.7, 95% CI: 1.2–2.4) among 6–15-year-old children and 44.4% (aOR: 1.9, 95% CI: 1.0-3.8) among adults 16-45 years old. Notably, the older age group (> 46 years) had a high aOR of 13.4 (95% CI: 3.3–54.7), indicating that this age group had the highest seroprevalence of CHIKV infection. One of the reported risk factors for acquiring chronic CHIKV disease is being more than 45 years old [48, 57]. A similar trend was observed among age groups in studies conducted in India, Singapore, and Nigeria [52, 56, 58]. The relationship between increasing age and CHIKV susceptibility has been attributed to the increased degree of exposure and impaired immune function in the elderly population [58, 59]. However, the seroprevalence of anti-CHIKV IgM antibodies was highest among children  $\leq$  5 years old and adults  $\geq$  46 years old, which is indicative of acute infection in the population. The  $\leq$  5-year-old and  $\geq$  46-year-old age groups are at the highest risk of severe disease due to CHIKV infection [46-50]. Infections in young children typically occur either through vertical transmission or through mosquito bites [60]. Interestingly, CHIKV seropositivity was not significantly associated with gender, as indicated by an adjusted OR of 1.0 (95% CI: 0.8–1.3). A study conducted in India reported a similar finding [52], although other studies showed significant differences in seropositivity by gender [56, 58].

The correlation between IgG and neutralizing antibodies (NAbs) (r = 0.76, 95% CI: 0.71-0.79) was significantly higher than that of IgM antibodies (r = 0.05, 95% CI: 0.002-0.1). Our findings agree with previous studies in which anti-CHIKV IgM antibodies have been shown to have weaker neutralizing effects than IgG [61]. Reportedly, ten days after disease onset, IgM plays a minimal role in overall neutralizing activity because neutralizing IgG becomes dominant [62]. This mechanism explains why our findings showed more IgG antibodies able to neutralize the virus and primarily indicated with a high titer.

The overall prevalence of NAbs in the study population was 32.5%. As observed with CHIKV seropositivity, the association of age and region with NAbs was statistically significant.

Similarly, the NAb prevalence rate increased with increasing age, as in CHIKV seropositivity. Additionally, by using an adjusted odds ratio, we determined that participants from Myeik (1.9; 95% CI: 1.2–2.9) and Yangon (2.4; 95% CI: 1.3–4.3) were approximately two times more likely to have anti-CHIKV neutralizing antibodies than those from Mandalay. Similarly, no significant difference was observed in the prevalence of NAbs by gender (p < 0.05). This finding indicates that a third of the study population is protected from infection, as other studies have shown that the presence of CHIKV NAbs with titers  $\geq$ 1:10 is correlated with protection from symptomatic infection and subclinical seroconversion [63–65].

The prevalence of CHIKV NAbs was higher in healthy volunteers (35.1%) than patients with febrile illness (28.1%). In contrast, the adjusted odds ratio revealed that CHIKV NAb prevalence was 1.6 times (95% CI: 1.2–2.2) higher in febrile patients than healthy volunteers. Furthermore, the adjusted AMEs estimates revealed that CHIKV NAb prevalence was 0.10% (95% CI: 0.03–0.16) higher in febrile patients than healthy volunteers. The higher proportion of CHIKV NAbs among healthy volunteers compared with febrile patients is attributed to the high number of DENV-seropositive individuals in the population. The samples from the febrile patients had been collected for DENV seroprevalence screening [36, 37]. 30.3% of the febrile patients had CHIKV antibodies, whereas 24.4% had both DENV and CHIKV antibodies. CHIKV and DENV normally co-circulate because they have similar vectors and are endemic in tropical regions [66]. The clinical symptoms observed in DENV- and CHIKV-positive individuals are initially similar; however, some distinguishing features are typically observed [66]. All of the clinical symptoms reported were observed in either CHIKV- or DENV-seropositive cases. However, 33.3% of arthralgia was noted in CHIKV-seropositive patients and 66.6% in combined CHIKV- and DENV-seropositive patients, but not in DENVonly cases. Arthralgia and rash have been identified as some of the typical clinical features of CHIKV although, in some cases, the infection can progress to debilitating polyarthritis that can last for months and years [67].

The findings in this study confirm the circulation of CHIKV in both healthy volunteers and febrile patients. However, this study had several limitations. The samples used in this work were collected previously for various studies, and this is apparent in the differing representation of each region. Additionally, the age groups were not evenly represented in the sampled regions. Lastly, the study was not able to show a systematic trend of infection between 2013 and 2018 because we lacked samples for 2014, 2016, and 2017. A more systematic seroprevalence study should be conducted to determine the extent of CHIKV infection among the population of Myanmar.

This study revealed the circulation of CHIKV in Mandalay, Yangon, and Myeik, which are located in northern Myanmar, southern Myanmar, and the extreme south of the country respectively; this indicates the presence of the virus in the country. The sample population had an overall IgG and IgM seropositivity rate of 34.5% (533/1,544) and a NAb prevalence rate of 32.5% (502/1,544). Additionally, our findings demonstrated the co-circulation of CHIKV with DENV antibodies in patients with febrile illness. This finding strengthens the need to incorporate screening for CHIKV during DENV outbreaks. With no current CHIKV treatment or vaccine, continuous monitoring of the virus through systematic surveillance is necessary. Lastly, the development of an affordable, reliable, and rapid diagnostic tool to detect multiple viruses that co-circulate is necessary for continuous seroprevalence surveillance in the region.

# Supporting information

**S1 Fig. Map of Myanmar showing the three study sites.** The regions marked in black are Mandalay, Yangon, and Myeik. Source: <u>https://aseanup.com/free-maps-myanmar/</u>. (TIF)

S2 Fig. Correlation between anti-CHIKV IgG/IgM antibodies and neutralizing antibodies (NAbs). Spearman's correlation coefficient *r* was used to determine the relationship between CHIKV IgG/IgM antibodies and NAbs. A) The red dotted horizontal and vertical lines represent the cutoff points for neutralization-positive ( $\geq$  10) and IgG-positive ( $\geq$  3000) samples, respectively. B) The red dotted horizontal and vertical lines represent the cutoff points for neutralization-positive (Positive–negative ratio  $\geq$  2) samples, respectively. *p* values < 0.05 were considered statistically significant. (TIF)

**S3 Fig. Seroprevalence of CHIKV and DENV among febrile patients.** The flow chart illustrates the seroprevalence rates of DENV, CHIKV, and DENV–CHIKV infections. (TIF)

**S4 Fig. Comparison of clinical symptoms among CHIKV- and DENV-infected patients.** The prevalence rate of the clinical presentation among the febrile patients is indicated on each bar. (TIF)

S1 Table. Validation of in-house anti-CHIKV IgM-capture ELISA with human anti-CHIKV Abcam IgM ELISA Kit (ab177848) as the standard. The sensitivity of the in-house anti-CHIKV IgM capture ELISA was 98.3% (95% CI: 90.9%–100%) and specificity was 88.0% (95% CI: 71.8%–96.6%), with an accuracy of 94.6%. (DOCX)

**S2 Table. Validation of the in-house anti-CHIKV IgG indirect ELISA with FRNT**<sub>50</sub> as the **standard.** The sensitivity of the in-house anti-CHIKV IgG indirect ELISA was 94.2% (95% CI: 88.9%–97.5%) and specificity was 100% (97.8%–100%), with an accuracy of 97.4%. (DOCX)

S3 Table. Model selection criteria for the association of CHIKV seropositivity with independent variables. Key: AIC, Akaike's Information Criterion; BIC, Bayesian Information Criterion. The logistic regression model with four independent variables was selected because it had the lowest AIC and BIC values. The model was correctly classified at 66.4% and the goodness of fit test was p = 0.3773. (DOCX)

DUCX)

S4 Table. Model selection criteria for the association of CHIKV neutralizing antibodies with independent variables. Key: AIC, Akaike's Information Criterion; BIC, Bayesian Information Criterion. The logistic regression model with the four independent variables was selected because it had the lowest AIC and BIC values. The model was correctly classified at 68.5%, and the goodness of fit test was p = 0.2289. (DOCX)

**S5 Table.** Average marginal effects (AMEs) estimates of CHIKV seroprevalence by age, region, gender, and health status. The *p*-values highlighted in bold indicate significant values. (DOCX)

**S6** Table. Average marginal effects (AMEs) estimates of CHIKV NAb prevalence by age, region, gender, and health status. The *p*-values highlighted in bold indicate significant values. (DOCX)

S7 Table. Anti-CHIKV neutralizing antibody (NAb) mean titers and geometric mean titers (GMT) categorized by age, region, gender, health status, and year. (DOCX)

#### Acknowledgments

The authors would like to thank all the study participants. The authors would also like to thank the members of the Department of Virology, Institute of Tropical Medicine, Nagasaki University for providing technical advice and support. The authors are grateful to Prof. Pandey Basu and Dr. Ando Tsuyoshi, for their unwavering support towards the accomplishment of this project. Elizabeth Ajema Chebichi Luvai is a recipient of the Japan International Cooperation Agency (JICA)'s Advanced Training Program for Fostering Global Leaders on Infectious Disease Control to Build Resilience against Public Health Emergencies, scholarship. Elizabeth Ajema Chebichi Luvai is grateful for the support from the Program for Nurturing Global Leaders in Tropical and Emerging Communicable Diseases of the Graduate School of Biomedical Sciences, Nagasaki University.

# **Author Contributions**

Conceptualization: Elizabeth Ajema Chebichi Luvai, Mya Myat Ngwe Tun.

Data curation: Elizabeth Ajema Chebichi Luvai, Nundu Sabiti Sabin, Mya Myat Ngwe Tun.

- Formal analysis: Elizabeth Ajema Chebichi Luvai, Nundu Sabiti Sabin, Mya Myat Ngwe Tun.
- **Investigation:** Elizabeth Ajema Chebichi Luvai, Aung Kyaw Kyaw, Saw Wut Hmone, Kyaw Zin Thant, Mya Myat Ngwe Tun.
- Methodology: Elizabeth Ajema Chebichi Luvai, Fuxun Yu, Shingo Inoue, Kouichi Morita, Mya Myat Ngwe Tun.

Supervision: Elizabeth Ajema Chebichi Luvai, Shingo Inoue, Kouichi Morita.

- Writing original draft: Elizabeth Ajema Chebichi Luvai.
- Writing review & editing: Elizabeth Ajema Chebichi Luvai, Aung Kyaw Kyaw, Nundu Sabiti Sabin, Kouichi Morita, Mya Myat Ngwe Tun.

#### References

- 1. Powers AM, Logue CH. Changing patterns of chikunya virus: Re-emergence of a zoonotic arbovirus. Journal of General Virology. 2007. https://doi.org/10.1099/vir.0.82858–0
- Cavrini F, Gaibani P, Pierro AM, Rossini G, Landini MP, Sambri V. Chikungunya: An emerging and spreading arthropod-borne viral disease. Journal of Infection in Developing Countries. 2009. <u>https://doi.org/10.3855/jidc.169</u> PMID: 20009275
- Caglioti C, Lalle E, Castilletti C, Carletti F, Capobianchi MR, Bordi L. Chikungunya virus infection: An overview. New Microbiologica. 2013. PMID: 23912863
- 4. Burt FJ, Rolph MS, Rulli NE, Mahalingam S, Heise MT. Chikungunya: A re-emerging virus. The Lancet. 2012. https://doi.org/10.1016/S0140-6736(11)60281-X PMID: 22100854
- Erin Staples J, Breiman RF, Powers AM. Chikungunya fever: An epidemiological review of a re-emerging infectious disease. Clinical Infectious Diseases. 2009. https://doi.org/10.1086/605496 PMID: 19663604
- Renault P, Solet JL, Sissoko D, Balleydier E, Larrieu S, Filleul L, et al. A major epidemic of chikungunya virus infection on Réunion Island, France, 2005–2006. Am J Trop Med Hyg. 2007. <u>https://doi.org/10.4269/ajtmh.2007.77.727</u>
- WHO. Introduction to the Regional Director's Annual Report on the Work of WHO in the South-East Asia Region covering the period 1 January-31 December 2014 (No. SEA/RC68/2). World Heal Organ. 2015. Available: https://apps.who.int/iris/handle/10665/182502
- Lumsden WHR. An epidemic of virus disease in Southern Province, Tanganyika territory, in 1952–1953 II. General description and epidemiology. Trans R Soc Trop Med Hyg. 1955. <u>https://doi.org/10.1016/0035-9203(55)90081-X</u>

- Sergon K, Njuguna C, Kalani R, Ofula V, Onyango C, Konongoi LS, et al. Seroprevalence of Chikungunya virus (CHIKV) infection on Lamu Island, Kenya, October 2004. Am J Trop Med Hyg. 2008. <u>https:// doi.org/10.4269/ajtmh.2008.78.333</u>
- Thiberville SD, Moyen N, Dupuis-Maguiraga L, Nougairede A, Gould EA, Roques P, et al. Chikungunya fever: Epidemiology, clinical syndrome, pathogenesis and therapy. Antiviral Research. 2013. <u>https:// doi.org/10.1016/j.antiviral.2013.06.009</u> PMID: 23811281
- Nunes MRT, Faria NR, de Vasconcelos JM, Golding N, Kraemer MUG, de Oliveira LF, et al. Emergence and potential for spread of Chikungunya virus in Brazil. BMC Med. 2015. <u>https://doi.org/10.1186/ s12916-015-0348-x PMID: 25976325</u>
- Tan Y, Pickett BE, Shrivastava S, Gresh L, Balmaseda A, Amedeo P, et al. Differing epidemiological dynamics of Chikungunya virus in the Americas during the 2014–2015 epidemic. PLoS Negl Trop Dis. 2018. https://doi.org/10.1371/journal.pntd.0006670 PMID: 30059496
- Rezza G, Nicoletti L, Angelini R, Romi R, Finarelli A, Panning M, et al. Infection with chikungunya virus in Italy: an outbreak in a temperate region. Lancet. 2007. <u>https://doi.org/10.1016/S0140-6736(07)</u> 61779–6
- Javelle E, Florescu SA, Asgeirsson H, Jmor S, Eperon G, Leshem E, et al. Increased risk of chikungunya infection in travellers to thailand during ongoing outbreak in tourist areas: Cases imported to europe and the middle east, early 2019. Eurosurveillance. 2019. <u>https://doi.org/10.2807/1560-7917.ES.2019</u>. 24.10.1900146 PMID: 30862335
- Díaz-Menéndez M, Esteban ET, Ujie M, Calleri G, Rothe C, Malvy D, et al. Travel-associated chikungunya acquired in Myanmar in 2019. Eurosurveillance. 2020. <u>https://doi.org/10.2807/1560-7917.ES.2020</u>. 25.1.1900721 PMID: 31937394
- Dudouet P, Gautret P, Larsen CS, Díaz-Menéndez M, Trigo E, von Sonnenburg F, et al. Chikungunya resurgence in the Maldives and risk for importation via tourists to Europe in 2019–2020: A GeoSentinel case series. Travel Med Infect Dis. 2020. https://doi.org/10.1016/j.tmaid.2020.101814 PMID: 32619732
- Calba C, Guerbois-Galla M, Franke F, Jeannin C, Auzet-Caillaud M, Grard G, et al. Preliminary report of an autochthonous chikungunya outbreak in France, July to September 2017. Eurosurveillance. 2017. https://doi.org/10.2807/1560-7917.ES.2017.22.39.17-00647 PMID: 29019313
- Riccardo F, Venturi G, Di Luca M, Del Manso M, Severini F, Andrianou X, et al. Secondary Autochthonous Outbreak of Chikungunya, Southern Italy, 2017. Emerg Infect Dis. 2019. <u>https://doi.org/10.3201/</u> eid2511.180949 PMID: 31625839
- Thaung U, Ming Swe CKT, Thein S. Epidemiological features of dengue and chikungunya infections in Burma. Southeast Asian J Trop Med Public Health. 1975; 6: 276–283. PMID: 126493
- Wimalasiri-Yapa BMCR, Stassen L, Huang X, Hafner LM, Hu W, Devine GJ, et al. Chikungunya virus in Asia–Pacific: a systematic review. Emerg Microbes Infect. 2019. https://doi.org/10.1080/22221751. 2018.1559708 PMID: 30866761
- Tun MMN, Inoue S, Thant KZ, Talemaitoga N, Aryati A, Dimaano EM, et al. Retrospective seroepidemiological study of chikungunya infection in South Asia, Southeast Asia and the Pacific region. Epidemiol Infect. 2016; 144: 2268–2275. https://doi.org/10.1017/S095026881600056X PMID: 27018566
- Ngwe Tun MM, Thant KZ, Inoue S, Nabeshima T, Aoki K, Kyaw AK, et al. Detection of East/Central/ South African genotype of chikungunya virus in Myanmar, 2010. Emerg Infect Dis. 2014. <u>https://doi.org/ 10.3201/eid2008.131431</u> PMID: 25062511
- Kyi Soe Aung Phay. Chikungunya reappears after 10 years. Myanmar Times. 2019. Available: <a href="https://www.mmtimes.com/news/chikungunya-reappears-after-10-years.html">https://www.mmtimes.com/news/chikungunya-reappears-after-10-years.html</a>
- Kyaw AK, Ngwe Tun MM, Nabeshima T, Soe AM, Mon KM, Thida, et al. Chikungunya virus infection in blood donors and patients during outbreak, Mandalay, Myanmar, 2019. Emerg Infect Dis. 2020. <u>https:// doi.org/10.3201/eid2611.201824</u> PMID: 33079056
- Pavri K. Disappearance of chikungunya virus from india and South East Asia. Transactions of the Royal Society of Tropical Medicine and Hygiene. 1986. <u>https://doi.org/10.1016/0035-9203(86)90358-5</u>(86) 90358–5 PMID: 3026069
- Carey Donald E. Chikungunya and dengue: A Case of Mistaken Identity? J Hist Med Allied Sci. 1971; XXVI: 243–62. Available: https://doi.org/https%3A//doi.org/10.1093/jhmas/XXVI.3.243 PMID: 4938938
- 27. Saswat T, Kumar A, Kumar S, Mamidi P, Muduli S, Debata NK, et al. High rates of co-infection of Dengue and Chikungunya virus in Odisha and Maharashtra, India during 2013. Infect Genet Evol. 2015. https://doi.org/10.1016/j.meegid.2015.08.006 PMID: 26247719
- Chang SF, Su CL, Shu PY, Yang CF, Liao TL, Cheng CH, et al. Concurrent isolation of chikungunya virus and dengue virus from a patient with coinfection resulting from a trip to Singapore. J Clin Microbiol. 2010. https://doi.org/10.1128/JCM.01228-10 PMID: 20881182

- Yang CF, Su CL, Hsu TC, Chang SF, Lin CC, Huang JC, et al. Imported chikungunya virus strains, Taiwan, 2006–2014. Emerg Infect Dis. 2016. https://doi.org/10.3201/eid2211.160404 PMID: 27767908
- Hapuarachchi HA, Bandara KB, Hapugoda MD, Williams S, Abeyewickreme W. Laboratory confirmation of dengue and chikungunya co-infection. Ceylon Med J. 2008. <u>https://doi.org/10.4038/cmj.v53i3</u>. 252 PMID: 18982804
- Nayar SK, Noridah O, Paranthaman V, Ranjit K, Norizah I, Chem YK, et al. Co-infection of dengue virus and chikungunya virus in two patients with acute febrile illness. Med J Malaysia. 2007; 62: 335–6. Available: http://www.ncbi.nlm.nih.gov/pubmed/18551940 PMID: 18551940
- 32. Chipwaza B, Mugasa JP, Selemani M, Amuri M, Mosha F, Ngatunga SD, et al. Dengue and Chikungunya Fever among Viral Diseases in Outpatient Febrile Children in Kilosa District Hospital, Tanzania. PLoS Negl Trop Dis. 2014. https://doi.org/10.1371/journal.pntd.0003335 PMID: 25412076
- Omarjee R, Prat CM, Flusin O, Boucau S, Tenebray B, Merle O, et al. Importance of case definition to monitor ongoing outbreak of chikungunya virus on a background of actively circulating dengue virus, St Martin, December 2013 TO January 2014. Eurosurveillance. 2014. <u>https://doi.org/10.2807/1560-7917</u>. ES2014.19.13.20753 PMID: 24721537
- Bagno FF, Figueiredo MM, Villarreal J, Pereira G de C, Godoi LC, da Fonseca FG. Undetected Chikungunya virus co-infections in a Brazilian region presenting hyper-endemic circulation of Dengue and Zika. J Clin Virol. 2019. https://doi.org/10.1016/j.jcv.2019.02.006 PMID: 30836281
- Sam IC, Chua CL, Chan YF. Chikungunya virus diagnosis in the developing world: A pressing need. Expert Review of Anti-Infective Therapy. 2011. https://doi.org/10.1586/eri.11.132 PMID: 22114956
- Kyaw AK, Ngwe Tun MM, Moi ML, Nabeshima T, Soe KT, Thwe SM, et al. Clinical, virological and epidemiological characterization of dengue outbreak in Myanmar, 2015. Epidemiol Infect. 2017. https://doi. org/10.1017/S0950268817000735 PMID: 28414004
- Ngwe Tun MM, Thant KZ, Inoue S, Kurosawa Y, Lwin YY, Lin S, et al. Serological characterization of dengue virus infections observed among dengue hemorrhagic fever/dengue shock syndrome cases in upper Myanmar. J Med Virol. 2013. https://doi.org/10.1002/jmv.23577 PMID: 23595687
- Kyaw AK, Ngwe Tun MM, Naing ST, Thida, Htwe TT, Mar TT, et al. Inapparent dengue virus infection among students in Mandalay, Myanmar. Trans R Soc Trop Med Hyg. 2020. https://doi.org/10.1093/ trstmh/trz071 PMID: 31638146
- Inoue S, Alonzo MTG, Kurosawa Y, Mapua CA, Reyes JD, Dimaano EM, et al. Evaluation of a dengue IgG indirect enzyme-linked immunosorbent assay and a japanese encephalitis IgG indirect enzymelinked immunosorbent assay for diagnosis of secondary dengue virus infection. Vector-Borne Zoonotic Dis. 2010. https://doi.org/10.1089/vbz.2008.0153 PMID: 19874185
- Pandey BD, Neupane B, Pandey K, Tun MMN, Morita K. Detection of chikungunya virus in Nepal. Am J Trop Med Hyg. 2015. https://doi.org/10.4269/ajtmh.15-0092 PMID: 26195462
- Kim YC, López-Camacho C, Garcia-Larragoiti N, Cano-Mendez A, Hernandez-Flores KG, Domínguez-Alemán CA, et al. Development of an E2 ELISA methodology to assess chikungunya seroprevalence in patients from an endemic region of Mexico. Viruses. 2019. <u>https://doi.org/10.3390/v11050407</u> PMID: 31052472
- 42. Rueda JC, Santos AM, Angarita JI, Giraldo RB, Saldarriaga EL, Ballesteros Muñoz JG, et al. Demographic and clinical characteristics of chikungunya patients from six Colombian cities, 2014–2015. Emerg Microbes Infect. 2019. https://doi.org/10.1080/22221751.2019.1678366 PMID: 31631794
- 43. Young J, Badgery-Parker T, Dobbins T, Jorgensen M, Gibbs P, Faragher I, et al. Comparison of ECOG/ WHO Performance Status and ASA Score as a Measure of Functional Status. J Pain Symptom Manage. 2015; 49: 258–264. https://doi.org/10.1016/j.jpainsymman.2014.06.006 PMID: 24996034
- 44. Williams. RK. Scalar measures of fit: Pseudo R2 and information measures (AIC & BIC). Univ Notre Dame. 2018;Notre Dame. Available: https://www3.nd.edu/~rwilliam/stats3/L05.pdf
- Dankers FJWM, Traverso A, Wee L van KS. Prediction Modeling Methodology. In: Kubben P, Dumontier M DA, editor. Fundamentals of Clinical Data Science. 2018. Available: <u>https://www.ncbi.nlm.nih.</u> gov/books/NBK543534/
- 46. Ruiz Silva M, Aguilar Briseño JA, Upasani V, van der Ende-Metselaar H, Smit JM, Rodenhuis-Zybert IA. Suppression of chikungunya virus replication and differential innate responses of human peripheral blood mononuclear cells during co-infection with dengue virus. PLoS Negl Trop Dis. 2017; 11: 1–15. https://doi.org/10.1371/journal.pntd.0005712 PMID: 28644900
- Lima Neto AS, Sousa GS, Nascimento OJ, Castro MC. Chikungunya-attributable deaths: A neglected outcome of a neglected disease. PLoS Negl Trop Dis. 2019; 13: 10–14. <u>https://doi.org/10.1371/journal.pntd.0007575</u> PMID: 31513593
- Castro APCR de, Lima RA, Nascimento J dos S. Chikungunya: vision of the pain clinician. Rev Dor. 2016; 17: 299–302. https://doi.org/10.5935/1806-0013.20160093

- Simarmata D, Chun D, Ng E, Kam Y-W, Lee B, Sia M, et al. Early clearance of Chikungunya virus in children is associated with a strong innate immune response OPEN. 2016. https://doi.org/10.1038/ srep26097 PMID: 27180811
- Sebastian MR, Lodha R, Kabra SK. Chikungunya infection in children. Indian J Pediatr. 2009; 76: 185– 189. https://doi.org/10.1007/s12098-009-0049-6 PMID: 19330307
- Vongpunsawad S, Intharasongkroh D, Thongmee T, Poovorawan Y. Seroprevalence of antibodies to dengue and chikungunya viruses in Thailand. 2017. <u>https://doi.org/10.1371/journal.pone.0180560</u> PMID: 28662144
- Kumar MS, Kamaraj P, Khan SA, Allam RR, Barde P V, Dwibedi B, et al. Seroprevalence of chikungunya virus infection in India, 2017: a cross-sectional population-based serosurvey. The Lancet Microbe. 2021; 2: e41–e47. https://doi.org/10.1016/s2666-5247(20)30175-0
- Somlor S, Vongpayloth K, Diancourt L, Buchy P, Duong V, Phonekeo D, et al. Chikungunya virus emergence in the Lao PDR, 2012–2013. 2017. <u>https://doi.org/10.1371/journal.pone.0189879</u> PMID: 29284012
- 54. Hossain MS, Hasan MM, Islam MS, Islam S, Mozaffor M, Khan MAS, et al. Chikungunya outbreak (2017) in Bangladesh: Clinical profile, economic impact and quality of life during the acute phase of the disease. PLoS Negl Trop Dis. 2018. https://doi.org/10.1371/journal.pntd.0006561 PMID: 29874242
- 55. Tun MMN, Kyaw AK, Hmone SW, Inoue S, Buerano CC, Soe AM, et al. Detection of zika virus infection in Myanmar. Am J Trop Med Hyg. 2018; 98: 868–871. <u>https://doi.org/10.4269/ajtmh.17-0708</u> PMID: 29363460
- 56. Omatola CA, Onoja BA, Fassan PK, Osaruyi SA, Iyeh M, Samuel MA, et al. Seroprevalence of chikungunya virus infection in five hospitals within Anyigba, Kogi State of Nigeria. Brazilian J Infect Dis. 2020. https://doi.org/10.1016/j.bjid.2020.01.001 PMID: 32001210
- Silva LA, Dermody TS. Chikungunya virus: Epidemiology, replication, disease mechanisms, and prospective intervention strategies. Journal of Clinical Investigation. 2017. pp. 737–749. <u>https://doi.org/10.1172/JCI84417 PMID: 28248203</u>
- Ang LW, Kam YW, Lin C, Krishnan PU, Tay J, Ng LC, et al. Seroprevalence of antibodies against chikungunya virus in Singapore resident adult population. PLoS Negl Trop Dis. 2017; 11: 1–13. <u>https://doi.org/10.1371/journal.pntd.0006163 PMID: 29281644</u>
- 59. Patil HP, Gosavi M, Mishra AC, Arankalle VA. Age-Dependent Evaluation of Immunoglobulin G Response after Chikungunya Virus Infection. Am J Trop Med Hyg. 2021; 1–6. <u>https://doi.org/10.4269/ajtmh.20-1504</u> PMID: 33432913
- Contopoulos-Ioannidis D, Newman-Lindsay S, Chow C, Labeaud AD. Mother-to-child transmission of Chikungunya virus: A systematic review and meta-analysis. 2018. <u>https://doi.org/10.1371/journal.pntd.</u> 0006510 PMID: 29897898
- Nitatpattana N, Kanjanopas K, Yoksan S, Satimai W, Vongba N, Langdatsuwan S, et al. Long-term persistence of Chikungunya virus neutralizing antibodies in human populations of North Eastern Thailand. Virol J. 2014; 11: 1–5. https://doi.org/10.1186/1743-422X-11-1 PMID: 24393133
- Chua CL, Sam IC, Chiam CW, Chan YF. The neutralizing role of IgM during early Chikungunya virus infection. PLoS One. 2017; 12: 1–16. https://doi.org/10.1371/journal.pone.0171989 PMID: 28182795
- Milligan GN, Schnierle BS, McAuley AJ, Beasley DWC. Defining a correlate of protection for chikungunya virus vaccines. Vaccine. 2019. https://doi.org/10.1016/j.vaccine.2018.10.033 PMID: 30448337
- Lum F-M, Teo T-H, Lee WWL, Kam Y-W, Rénia L, Ng LFP. An Essential Role of Antibodies in the Control of Chikungunya Virus Infection. J Immunol. 2013. <u>https://doi.org/10.4049/jimmunol.1300304</u> PMID: 23670192
- 65. Yoon IK, Srikiatkhachorn A, Alera MT, Fernandez S, Cummings DAT, Salje H. Pre-existing chikungunya virus neutralizing antibodies correlate with risk of symptomatic infection and subclinical seroconversion in a Philippine cohort. Int J Infect Dis. 2020; 95: 167–173. <u>https://doi.org/10.1016/j.ijid.2020.03.073</u> PMID: 32247051
- Mohd Zim MA, Sam IC, Omar SFS, Chan YF, AbuBakar S, Kamarulzaman A. Chikungunya infection in Malaysia: Comparison with dengue infection in adults and predictors of persistent arthralgia. J Clin Virol. 2013. https://doi.org/10.1016/j.jcv.2012.10.019 PMID: 23201456
- Shahid U, Farooqi JQ, Barr KL, Faisal S, Id M, Jamil B, et al. Comparison of clinical presentation and out-comes of Chikungunya and Dengue virus infections in patients with acute undifferentiated febrile illness from the Sindh region of Pakistan. 2020. <u>https://doi.org/10.1371/journal.pntd.0008086</u> PMID: 32203509